Author's response to reviews

Title: A feasibility study of expert patient and community mental health team led bipolar psychoeducation groups: implementing an evidence based practice.

Authors:

- Katharine Coulthard (katharine.coulthard@ntw.ac.uk)
- Dipty Patel (dipty.patel@ntw.nhs.uk)
- Clare Brizzolara (clare.brizzolara@sunderland.ac.uk)
- Richard Morriss (richard.morriss@nottingham.ac.uk)
- Stuart Watson (stuart.watson@ncl.ac.uk)

Version: 2 Date: 15 October 2013

Author's response to reviews:

see letter